

Assessment of consensus forecasts accuracy

Novotný, Filip; Raková, Marie 2010 Dostupný z http://www.nusl.cz/ntk/nusl-124038

Dílo je chráněno podle autorského zákona č. 121/2000 Sb.

Tento dokument byl stažen z Národního úložiště šedé literatury (NUŠL). Datum stažení: 17.05.2024

Další dokumenty můžete najít prostřednictvím vyhledávacího rozhraní nusl.cz .

WORKING PAPER SERIES 14

Filip Novotný and Marie Raková: Assessment of Consensus Forecasts Accuracy: The Czech National Bank Perspective



WORKING PAPER SERIES

Assessment of Consensus Forecasts Accuracy: The Czech National Bank Perspective

> Filip Novotný Marie Raková

> > 14/2010

CNB WORKING PAPER SERIES

The Working Paper Series of the Czech National Bank (CNB) is intended to disseminate the results of the CNB's research projects as well as the other research activities of both the staff of the CNB and collaborating outside contributor, including invited speakers. The Series aims to present original research contributions relevant to central banks. It is refereed internationally. The referee process is managed by the CNB Research Department. The working papers are circulated to stimulate discussion. The views expressed are those of the authors and do not necessarily reflect the official views of the CNB.

Distributed by the Czech National Bank. Available at http://www.cnb.cz.

Reviewed by: Jiří Slačálek (European Central Bank) Marcus Kappler (ZEW Mannheim) Jan Babecký (Czech National Bank)

Project Coordinator: Michal Franta

© Czech National Bank, December 2010 Filip Novotný, Marie Raková

Assessment of Consensus Forecasts Accuracy: The Czech National Bank Perspective

Filip Novotný and Marie Raková*

Abstract

Consensus Economics forecasts for euro-area GDP growth, consumer and producer price inflation and the USD/EUR exchange rate are used by the Czech National Bank to make assumptions about future external economic developments. This paper compares the accuracy of the aforementioned Consensus forecasts to those of the European Commission, International Monetary Fund and Organization for Economic Co-operation and Development, and also to the naïve forecast and the forecast implied by the forward exchange rate. In the period from 1994 to 2009 Consensus forecasts for effective euro-area consumer price inflation and GDP growth beat the alternatives by a difference which is typically statistically significant. The results are more diverse for the pre-crisis sample (1994–2007). The Consensus forecast for euro-area producer price inflation significantly outperforms the naïve forecast in the short-term. Finally, the Consensus forecast for the USD/EUR exchange rate during the period from 2002 to 2009 is more precise than the naïve forecast implied by the forward rate.

JEL Codes: E37, E58.

Keywords: Forecasting accuracy, prediction process, survey forecasts.

^{*} Filip Novotný, corresponding author, Czech National Bank, External Economic Relations Division, Na Příkopě 28, Prague 1, 11503, Czech Republic (e-mail: filip.novotny@cnb.cz).

Marie Raková, Czech National Bank, External Economic Relations Division, Na Příkopě 28, Prague 1, 11503, Czech Republic (e-mail: Marie.Rakova@centrum.cz).

We have benefited from valuable comments from the Monetary and Statistics Department of the Czech National Bank on previous versions of this paper as well as comments by the referees of the paper and participants at the 2010 Czech Economic Society Conference. In addition, we are grateful to David Kocourek for his technical assistance. Nevertheless, all errors and omissions are ours. The views expressed in this paper are those of the authors and do not necessarily represent those of the Czech National Bank.

Nontechnical Summary

In this paper we compare the accuracy of the Consensus Economics forecasts to those of the European Commission, International Monetary Fund and Organization for Economic Cooperation and Development, and also to the naïve forecast and forecast implied by forward exchange rates. The Consensus Economics forecasts are used by the Czech National Bank to make assumptions about future external economic developments.

We make the comparison for the period from 1994 to 2009 and for the following variables: GDP growth, consumer and producer price inflation and the USD/EUR exchange rate. The forecasts for each macroeconomic variable are assessed in terms of the effective indicator, where each country forecast is weighted by the respective country share in Czech exports. The same effective indicators are applied in the CNB's prediction process. The comparable effective Consensus, EC, IMF and OECD forecasts are constructed before the forecasts are evaluated.

The following standard measures are used for the descriptive analysis of the forecasting accuracy: mean forecast error (MFE), mean absolute forecast error (MAFE), mean absolute percentage error (MAPE), mean squared error (MSE), and root mean squared error (RMSE). We also perform a forecast comparison regression and, finally, the Diebold-Mariano test of statistical significance in the forecasting errors of competing forecasts is performed (Diebold and Mariano, 1995). In addition to this, the comparison is provided with a visual forecast assessment of the last three years of our sample.

Forecasting accuracy is compared before and during the recent financial crisis. With respect to the GDP growth and CPI inflation forecasts, we find that the Consensus Economics forecasts usually outperform the others significantly, especially during the whole period from 1994 to 2009. The results are not that strong for the pre-crisis period. The Consensus forecasts beat the international institutions' forecasts mainly for current-year forecasts and, additionally, Consensus is superior to the naïve forecasts at all forecast horizons.

In accordance with the previous literature on this topic, we confirm a relatively low accuracy for next-year GDP growth forecasts, which are biased upwards by all institutions, whereas the CPI inflation forecasts are unbiased.

The PPI inflation and USD/EUR forecasts are compared only with the naïve forecast and the forward rate forecast in the case of USD/EUR due to the absence of comparable forecasts by international institutions. The Consensus PPI inflation forecast significantly beats the naïve forecast for the current year. The Consensus forecast for USD/EUR improved dramatically after 2002, when it significantly outperforms the naïve forecast for all forecasting horizons and the forward implied forecast for the one-year horizon. A stronger dollar was systematically predicted over the observed period by all the assessed forecasts.

1. Introduction

The monetary policy regime of direct inflation targeting was adopted by the Czech National Bank (CNB) in 1998. Under inflation targeting, the forecast for consumer price inflation (CPI) at the monetary policy horizon is of great relevance to the decision-making on the current interest rates. An important assumption of the CNB's macroeconomic forecast is the external economic outlook, which is partly derived from the Consensus Forecasts (hereinafter "Consensus").

Consensus is a regular monthly survey publication which provides forecasts and views on the principal macroeconomic indicators, including GDP growth, price inflation, interest rates and exchange rates, in over 85 countries. The surveys mainly contain the expectations of the main investment banks and other well-known analytical centers. Consensus serves as a background for the decision-making processes of economic agents, including central banks. It is therefore relevant to research the ex-post accuracy of Consensus, because if we are aware of the historical performance of the Consensus forecasts we are then in a better position to judge the information it provides for our future decision making.

In this paper, we decided to compare the forecasting accuracy of Consensus with the corresponding forecasts of the International Monetary Fund (IMF), Organization for Economic Co-operation and Development (OECD), and European Commission (EC) and also with the naïve forecast and forecasts implied by the forward exchange rate. Although the Consensus forecasts were already assessed at the CNB on an *ad hoc* basis, this paper provides a comprehensive assessment of Consensus. The forecasts of real GDP growth, CPI inflation and PPI inflation for Austria, France, Germany and Italy are assessed along with the nominal USD/EUR exchange rate forecasts. Contrary to the available literature, we assess the forecasts for each macroeconomic variable in terms of the effective indicator, where each country forecast is weighted by the respective country share in Czech exports. The same effective indicators are applied in the CNB's prediction process.

Our main focus is on the Consensus performance in the period from 1994 to 2009, but we also separately assess the pre-crisis period from 1994 to 2007 to reveal any potential impacts which the financial crisis might have imposed on the overall forecasting accuracy. In addition, we assess the accuracy of the forecasts for a shorter period starting in 2002. Standard descriptive statistics of the accuracy of the forecasts as well as statistical tests for significance in the forecasting errors are applied.

The paper is organized as follows. The next section describes the previous literature on Consensus evaluation. Section 3 provides the characteristics of Consensus and its application in the CNB's prediction process. Section 4 deals with the data and methodology description. Section 5 covers the empirical results for three macroeconomic variables: CPI inflation, PPI inflation and real GDP growth. Section 6 proceeds with the results for the USD/EUR exchange rate, and Section 7 summarizes the main findings.

2. Related Literature

Because of the rather long history and wide acceptance of Consensus, there are a vast number of publications dealing with its accuracy. Similarly to our paper, Batchelor (2001, 2007) compares the Consensus forecasts with the corresponding forecasts published by the IMF and OECD. According to him, the Consensus forecasts are less biased and more accurate for the G7 countries. The GDP growth forecasts have been overestimated (higher than actual values) in the cases of France, Germany, Italy and Japan. Conversely, the CPI inflation forecasts have been unbiased. Similarly, Osterloh (2008), who dealt solely with the Consensus forecasts for German real GDP growth in the period from 1995 to 2005, shows that forecasters pooled by Consensus were systematically overestimating the growth rates. In addition, he discovers a relatively low accuracy for next-year forecasts compared to a simple naïve forecast. The Consensus forecasts for 12 industrial countries over the period from 1996 to 2006 were also explored regarding their bias and information efficiency in Ager et al. (2009). The authors showed that the forecasts for some countries, e.g., for Germany and Italy, and in particular the forecasts for horizons longer than one year, were systematically biased. Moreover, forecast information efficiency had to be rejected in almost all cases.

Among others, Timmermann (2006) analyzed the IMF forecasts in comparison with Consensus. He finds the IMF forecasts similar to those of Consensus and identifies weaknesses in the IMF forecasts. Bowle et al. (2007) assess the Survey of Professional Forecasters (SPF) in the period from 1999 to 2006. The SPF surveys approximately 75 forecasters from the European Union and asks them for their short and long-term expectations about euro-area economic variables – inflation, GDP growth and unemployment. Bowle et al. (2007) conclude that the SPF provides better forecasts for the GDP growth of the euro area in comparison with Consensus. Nevertheless, their results are not surprising given that the SPF focuses solely on the euro area and has the advantage of a larger sample of respondents (some of whom contribute to Consensus as well).

A different approach was applied by Dovern et al. (2009). Contrary to the previous literature on the topic, the authors did not concentrate on mean Consensus, but analyzed the dispersion (heterogeneity) of individual Consensus forecasters. They found that disagreement among forecasters tends to rise during recessions and is particularly pronounced in the case of real variables (GDP growth, consumption growth, investment growth and the unemployment rate). In addition, there is a downward trend in the disagreement of the forecasts for nominal variables (CPI inflation and interest rates) and this is lower in the case of countries with an independent central bank (for 35 percent)¹. It is also mentioned that Consensus is sensitive to current conditions, due to a strong correlation of the one-year-ahead forecasts with the current actual values.

More generally, Ang et al. (2007) compared four different methods of inflation forecasting and they assert that survey forecasts (the Livingston, Michigan, and SPF surveys) outperform the other three methods, namely, the ARIMA model, the economic model of the Phillips curve and the term structure of interest rates.

¹ Countries with an independent central bank during the whole period under review (1989–2006) comprise Canada, France, Germany and the United States. Countries which did not have an independent central bank for the whole period are Japan and the United Kingdom.

With respect to the Czech National Bank's own inflation forecasts, Antal et al. (2008) analyzed the bias of the forecasts in relation to any undershooting of the inflation target. They conclude that the inflation forecasting error was decreasing over time. About half of the apparent target undershooting in 2003 was due to errors in the forecasts of exogenous variables (foreign interest rates, GDP, and inflation). Nevertheless, in other years, errors in the exogenous variable forecasts did not contribute significantly to target undershooting. In addition, Babecký and Podpiera (2011) compared the accuracy of the CNB's inflation forecasts with other financial institutions' forecasts for the Czech Republic.

3. Consensus Forecasts Application at the Czech National Bank

Consensus is a regular monthly publication of the London-based Consensus Economics (http://www.consensuseconomics.com), which was founded in 1989. Consensus Economics is a private research organization that pools more than 700 economic analysts and economic research centers, mostly from private investment banks. Consensus does not pool any central bank, national government or international institution such as the European Bank for Reconstruction and Development (EBRD), EC, IMF or OECD. The number of participating analytical centers varies across individual countries and also for different time periods for a given country. It oscillates mostly between 10 and 30 analytical centers for a certain economic variable and country.

Initially, Consensus focused only on the G7 economies, but it has gradually increased its range of countries and currently comprises forecasts for more than 85 countries, which are surveyed in separate regional publications (see Table 1).

Consensus Publication	Surveyed Countries/Commodities	Forecasted Variables	Number of Participating Analytical Centers	
Consensus Forecasts (G7 & Western Europe)	G7 countries plus the Netherlands, Norway, Spain, Sweden, Switzerland and other countries ^{a)}	Various indicators for each country include altogether: - real GDP, - private consumption,	From 8 for the Netherlands to 31 for the euro area	
Eastern Europe Consensus Forecasts	stern Europe Consensus 14 countries of Eastern Europe ^{b)} plus a limited amount of economic indicators for other 13 countries 15 Asign countries		From 7 for Lithuania to 16 for Hungary, Poland or Russia ^{c)}	
Asia Pacific Consensus Forecasts	15 Asian countries, Australia and New Zealand	 - exports and imports of goods and services, - current account balance. 	From 11 for Indonesia to 21 for Japan	
Latin American Consensus Forecasts	14 Latin American countries	 government budget balance, 3M money market rates, 10Y government bond 	From 12 for Venezuela to 23 for Argentina	
Consensus Forecasts (USA)	USA ^{d)}	yields	22	
Foreign Exchange Consensus Forecasts	About 90 currency pa	irs, including CZK/EUR	Up to 90 for USD/EUR or YEN/USD	
Energy and Metals Consensus Forecasts (quarterly)	Crude oil, gasoline, natural aluminum, copper, nickel, gold, silver, platinum, palla	l gas, coal, uranium, lead, zinc, steel, iron ore, adium	30	

Table 1: List of Consensus Publications

Note: ^{a)} Austria, Belgium, Denmark, Egypt, Finland, Greece, Ireland, Israel, Nigeria, Portugal, Saudi Arabia and South Africa. Only GDP, industrial production, consumer prices and the current account balance are forecasted for these countries.

^{b)} Bulgaria, Croatia, Czech Republic, Estonia, Hungary, Latvia, Lithuania, Poland, Romania, Russia, Slovakia, Slovenia, Turkey and Ukraine.

^{c)} Roughly 15 participating analytical centers for the Czech Republic (number varies over time).

^{d)} 20 economic and financial indicators for the USA and 3–4 indicators for other selected countries. *Source*: Consensus Economics

(http://www.consensuseconomics.com/Economic_Forecast_Publications.htm).

The final Consensus forecast for a given country and a certain economic variable is a simple average of the forecasts provided by each participating forecaster. The average (consensual) forecast is published together with its standard deviation, with the forecasts of individual respondents being made available. Nevertheless, in this paper, we will focus exclusively on mean Consensus assessment because the mean forecasts are used by the CNB. Moreover, even if we identify a forecaster who systematically outperforms the other Consensus forecasters, it will be difficult for the CNB to refer to this particular forecaster only. Additionally, the general advantage of the mean forecast is that it eliminates (to a certain degree) any possible systematic errors in individual forecasts.

Consensus contains both fixed-event forecasts and rolling-event forecasts. Fixed-event forecasts refer to a certain time (calendar year) in the future. Specifically, forecasts are prepared for the current and the next calendar year similarly to the EC, IMF and OECD forecasts. Consensus also

publishes long-term forecasts twice a year. Long-term forecasts comprise yearly outlooks six years ahead and the average outlook for the next five years. The forecast horizon therefore adds up to 11 years in total.

Fixed-event forecasts imply a changing forecast horizon, i.e., the forecast horizon becomes shorter as we approach our target (forecast) year. This type of forecast is used, among other things, for CPI and PPI inflation and GDP growth.

In the case of rolling-event forecasts, the length of the forecast horizon is steady over time. Consensus rolling-event forecasts are for three-month, one-year and two-year horizons. This type of forecast is used, e.g., for exchange rates, interest rates and the WTI oil price.

The Consensus forecasts for selected external economic variables started to be used at the CNB in the second half of 2002. This was connected with the introduction of the bank's unconditional quarterly projection model (QPM) of the Czech economy, which was replaced by the new dynamic structural model (g3) in 2008, and with the resulting need for a consistent forecast of external (foreign) economic variables.

Variable	Unit	Source of Forecast						
	Variabi	les from Consensus						
Effective euro-area industrial producer prices (PPI)	Weighted average of harmonized indices for 14 euro-area countries based on seasonally adjusted series. Forecasts for each quarter are derived from the Consensus whole-year forecasts							
Effective euro-area consumer prices (CPI)	y-o-y (in percent)	Weighted average of non-harmonized indices for 14 euro-are countries based on seasonally adjusted series. Forecasts fo						
Effective euro-area GDP	y-o-y (in percent)	each quarter are derived from the Consensus whole-year forecasts						
Nominal euro exchange rate	USD/EUR	Three-months and one-year-ahead point forecasts (interpolated)						
Varia	bles deriv	ed from market derivatives						
Brent oil price	USD/barrel	rel Outlook is derived from futures contracts as of the Consens survey date						
Gasoline price USD/t Outlook is derived from swap contracts as of survey date								
3M Euribor	percent	Outlook is derived from market interest rates (FRA)						

 Table 2: List of Variables Covered in Reference Scenarios of External Factors

The CNB subscribes to the Consensus Forecasts (G7 & Western Europe), Eastern Europe Consensus Forecasts and Foreign Exchange Consensus Forecasts. Eight of a total of twelve monthly Consensus issues are used in the CNB's prediction process. This number corresponds to the eight monetary policy meetings held during a calendar year. Nevertheless, for internal purposes, outlooks for the external economic environment are updated continuously.

The bank's external economic outlooks use only a limited number of the economic variables which are provided by Consensus. Moreover, the original Consensus forecasts are adjusted to make them appropriate for the bank's prediction process. Specifically, the forecasts for CPI and PPI inflation and GDP growth, which are available as a whole-year percentage changes, are equally decomposed into individual quarters in order to be suitable for the quarterly prediction model. Subsequently, the effective euro-area indicator is calculated by weighting each euro-area country by its share in total Czech exports.

The effective euro-area indicator for the Czech Republic comprises 14 euro-area countries, with only Luxemburg and Malta not being included. Germany (47 percent), Slovakia (14 percent) and France (8 percent) are characterized by having the largest shares in the effective indicator.

The adjustment procedure is easier in the case of the USD/EUR exchange rate, where the Consensus point forecasts (rolling-event forecast) three months and one year ahead are interpolated.

In parallel with Consensus the remaining external economic outlooks for the oil price, gasoline price and three month Euribor are derived from market instruments (derivatives). The market implied outlooks are prepared as of the Consensus survey date, i.e., on the second Monday of each calendar month, in order to preserve the time consistency of all the predicted variables. Finally, the outlooks for each variable are put together and presented in an internal CNB document called the Reference Scenarios of External Factors (see Table 2 for an overview).

4. Data and Methodology

The forecasting accuracy of the Consensus forecasts for real GDP growth and CPI inflation is compared with corresponding forecasts published by international institutions – EC, IMF and OECD. In addition, the Consensus forecasts are assessed against the naïve forecasts. The forecasts for PPI inflation and the USD/EUR exchange rate are assessed, due to a lack of alternative forecasts, against the naïve forecast only; moreover, the forecast for USD/EUR is assessed also against the forecast derived from the forward exchange rate. As mentioned previously, the forecast horizon for CPI and PPI inflation and real GDP growth is the current year and the next year. The forecast for the USD/EUR exchange rate is available at three-month, one-year and two-year horizons.

Our naïve forecast is a random walk, AR(1) process, with the coefficient 1. The naïve forecast is thus the last-year growth rate of GDP, CPI inflation and PPI inflation, which is simply prolonged to the current year and the next year. We proceed similarly with the naïve forecast for the USD/EUR exchange rate. The actual value of the exchange rate as of the Consensus survey date is used as a constant naïve forecast at three-month, one-year and two-year horizons. The naïve forecasts used in the paper simulate the behavior of a naïve forecaster who mechanically projects (without any additional judgment) the last available data into the future. Alternatively, ARIMA forecasts, which are another commonly used benchmark based on past values, would probably provide better forecasts. We therefore deem our benchmark as having the worst available expectation about the future, which is our objective.

We encountered the problem of different publication dates of the compared forecasts. The main difference between Consensus and the international institutions' forecasts was the frequency at which the forecasts are published. The advantage of Consensus is its monthly frequency, while the EC, IMF and OECD publish their standard forecasts twice a year. The EC and the OECD publish their forecasts every May and November, whereas the IMF forecasts are published one month ahead, i.e., in April and October. Nevertheless, in 2007 the international institutions started to publish interim forecasts as well and the forecasting frequency has thus doubled.

In order to provide an evaluation of the Consensus and international institutions' forecasts, only the Consensus issues corresponding to the month of the international institutions forecasts' releases were used for relative forecast assessment. Accordingly, the IMF forecasts were compared with the April and October Consensus issues, and the EC and OECD forecasts with the May and November Consensus issues.

In addition, as mentioned previously, the real GDP growth and CPI and PPI inflation forecasts were assessed in terms of the effective indicator and not as a single forecast for a given country. This approach reflects the procedure used by the CNB whereby external economic developments are proxied by effective indicators where each country variable is weighted by the country share in total Czech exports. The comparable effective Consensus, EC, IMF and OECD forecasts are constructed before the forecasts are evaluated.

The effective indicator forecast can be expressed as follows:

$$f_h^{eff} = \sum_{i=1}^n f_h^i w_i , \qquad (1)$$

where f_h^{eff} is the effective indicator forecast at horizon *h* (*h*=1 for the current year and *h*=2 for the next year)_{*n*} f_h^i is a particular country forecast at horizon *h*, and w_i is the country share in Czech exports ($\sum_{i=1}^{n} w_i = 1$).

Only Austria, France, Germany and Italy are covered in our effective indicator. These four countries are used because previous forecasts are not available for every euro-area country. Nevertheless, this is a sufficient approximation given that Austria, France, Germany and Italy account for approximately 70 percent of total Czech exports.

The weight of Germany in our restricted effective indicator is 67 percent. France is weighted by 12 percent, and Italy and Austria by 11 and 10 percent, respectively. We calculate both the effective actual time series of real GDP growth and CPI and PPI inflation, and also their corresponding effective forecasts for each institution, including the naïve forecasts.

The forecasts are assessed for the period from 1994 to 2009, with forecast accuracy also being assessed for the shorter period from 2002 to 2009. The motivation for the shorter period is that Consensus was implemented at the CNB in 2002. The forecast for the USD/EUR exchange rate is assessed for the periods from 1999 to 2009 and from 2002 to 2009. In addition, the forecasts are assessed separately in the pre-crisis period (excluding 2008 and 2009) to reveal the potential effects of the financial crisis on overall forecasting accuracy (see Annex 1-4 where the results for the pre-crisis period are shown).

In order to assess forecasting accuracy, the forecasting errors are calculated first:

$$e_t = a_t - f_t \,, \tag{2}$$

where a_t represents the actual (realized) value and f_t is its corresponding forecast value. Similarly to Batchelor (2001) and Osterloh (2008), we use unrevised actual values by taking the actual value of the previous year from the current year Consensus June issue. The actual value of GDP growth in 1994 is thus taken from the June 1995 issue of Consensus. We proceed uniformly in the following years.

Unrevised actual data is used because economic forecasters are assumed to be quite unlikely to anticipate the extent of data revisions. It is more likely that forecasters will make their forecasts anticipating the same methods of data construction as those used by governmental statistical agencies. Our approach is the opposite to that used, for example, by Croushore (2010), who based his analysis solely on the latest available time series of actual data, which naturally contain *ex post* revisions.

If we look at the actual time series of effective GDP growth, which is probably the most affected by data revisions, we find that the difference between our method and the actual time series downloaded at the end of 2010 is, on average, zero. Nevertheless, there are differences in individual years. The largest negative deviation (-0.6 percentage points) was observed in 2006 and the largest positive deviation (0.5 percentage points) in 2004. The standard deviation over the whole sample is 0.3 percentage points.

The following standard measures are considered for the descriptive analysis of forecasting accuracy:

Mean forecast error (MFE), which is the average deviation between the actual value and its corresponding forecast. A positive (negative) MFE value means that forecast values are on average underestimated (overestimated). The disadvantage of MFE is that this measure averages positive and negative deviations (adverse forecast errors are eliminated).

$$MFE = \frac{1}{T} \sum_{i=1}^{T} e_i$$
(3)

Mean absolute forecast error (MAFE), which calculates the average errors in absolute terms.

$$MAFE = \frac{1}{T} \sum_{t=1}^{T} |e_t|$$
(4)

Mean absolute percentage error (MAPE)², which, in addition to MAFE, represents the forecast errors as a percentage of the actual values. Due to this characteristic, economic variables which are different in level, e.g., GDP growth and CPI inflation, can be compared more appropriately Nevertheless, one can argue that MAPE, as an indicator for the comparison of different forecasts, does not take into account another characteristic of economic variables, namely, their volatility (more stable economic variables can be more easily predicted). We try to incorporate time series volatility partly in Section 5.3 by introducing a combined forecast.

 $^{^{2}}$ MAPE is used in the *ad hoc* graphical analysis, where the forecast accuracy of different economic variables is presented (only for Consensus forecasts and for the comparison between the Consensus and the naïve forecasts).

$$MAPE = \frac{\frac{1}{T} \sum_{t=1}^{T} |e_t|}{\frac{1}{T} \sum_{t=1}^{T} a_t}$$
(5)

Mean squared error (MSE) measures the average of the squared forecasting errors. The advantage of MSE is that larger forecasting errors are penalized more.

$$MSE = \frac{1}{T} \sum_{t=1}^{T} e_t^2 \tag{6}$$

Root mean squared error (RMSE) is, *de facto*, MSE which is expressed in the same units as the assessed time series.

$$RMSE = \sqrt{\frac{1}{T} \sum_{t=1}^{T} e_t^2}$$
(7)

A forecast is considered to be biased if it is systematically too high or too low. As in, e.g., Ager et al. (2009) and Croushore (2010), we verify the forecast bias by regressing the forecast errors on a constant:

$$e_t = \alpha + \varepsilon_t. \tag{8}$$

The coefficient α represents MFE, so that equation (8) is effectively a test for a zero mean of the forecasting error. The null hypothesis, that the forecasts are unbiased, would hold if $\alpha = 0$. Later in the paper we also use two modifications of equation 8 for the absolute $|e_t|$ and squared e_t^2 forecasting errors.

In addition, we perform a forecast comparison regression, following Ang et al. (2007) and Stock and Watson (1999):

$$a_t = \lambda f_t^{cf} + (1 - \lambda) f_t^{alt} + u_t, \qquad (9)$$

where f_t^{cf} is the Consensus forecast, f_t^{alt} represents the alternative forecast, i.e., the international institutions' forecasts and the naïve forecast, λ is the corresponding coefficient, and u_t is the forecast error associated with a combined forecast. If $\lambda=0$, then the Consensus forecast adds nothing to the alternative forecast, and we thus conclude that the alternative forecast outperforms the Consensus benchmark. If $\lambda=1$, then, conversely, the alternative forecast adds nothing to the Consensus forecast. It is possible that if λ is significantly negative then it does contain information but of a perverse kind. In this specific example, when the Consensus forecast is raised (lowered), the optimal combined forecast should be reduced (increased).

Finally, the Diebold-Mariano (D-M) test of statistical significance in the forecasting errors of competing forecasts is used (Diebold and Mariano, 1995). The D-M test assesses the quality of each forecast using a loss function of the forecast error. It is common to use the MSE loss differential to evaluate business cycle forecasts:

$$\overline{d} = MSE^{cf} - MSE^{alt} = 0, \qquad (10)$$

where MSE^{cf} and MSE^{alt} are the mean squared errors of the Consensus and alternative forecasts, respectively. The null hypothesis (H_0) that two forecasts are on average the same is tested.

The D-M test statistic is expressed in the following form:

$$DM = \bar{d} / [2\pi \hat{g}_d(0) / T]^{1/2}, \qquad (11)$$

where \overline{d} is the average difference between the errors of two forecasts at time t = 1, 2, 3, ..., T and $\hat{g}_d(0)$ is a consistent estimation of a distribution function $g_d(0)$.

5. Empirical Results for Effective GDP Growth and CPI and PPI Inflation

5.1 Visual Forecast Assessment

Firstly, we provide a visual forecast assessment of the last three years of our sample. We find the initial visual inspection of these particular years helpful before we proceed to the empirical results.

The evolution of the effective GDP growth forecasts is illustrated in Figure 1 for 2007 (see Figure 1A) and 2008 (see Figure 1B). Both figures show the evolution of the forecasts during the 24 months before the realization of the predicted value. Consensus, due to its high frequency of forecasts, publishes 24 forecasts for a given variable over the period (twelve monthly forecasts during a calendar year). The international institutions publish their standard forecasts twice a year, in contrast to Consensus.



Figure 1: Effective Euro-Area GDP Forecasts – Visual Comparison

Note: Dashed lines depict the interval containing 68 percent of the individual forecasters pooled by Consensus.

In Figure 1A, month 24 on the *x* axis represents the first available Consensus GDP forecast for 2007 (we exclude the long-term forecasts). The forecast was published in the January 2006 Consensus issue. The first corresponding IMF, EC and OECD forecasts were released in April and May 2006, respectively (see the points at 21 and 20 months). Finally, the straight line represents the *ex-post* known actual values of GDP growth in 2007 and 2008, respectively.³ At first sight, it is obvious that the forecasts for both years were relatively distant from the actual value at the beginning of the forecast horizon. The forecasts approached the actual values only in the last months of the horizon.

The intervals of individual Consensus forecasters are illustrated by the dashed lines, which are given by one standard error of the individual Consensus forecasts. It is apparent that a gradual shortening of the horizon implies a narrowing of the intervals. In other words, the individual Consensus forecasts draw nearer to each other over time, which we interpret as meaning decreasing uncertainty of the mean Consensus forecast.

Nevertheless, exceptions occur in turbulent times, when the uncertainty of the mean Consensus forecast might increase despite a shortening of the forecast horizon (for a detailed analysis see also Dovern et al., 2009). A widening of the interval of the individual Consensus forecasts for 2008 effective euro-area GDP growth was observed in September 2007 (16 months before the end of the forecast horizon) and again in August and in November 2008 (5 and 2 months before the end of the forecast horizon). A temporary improvement of the GDP growth forecast was observed in June 2008 (see Figure 1B).



Figure 2: Effective Euro-Area CPI Forecasts – Visual Comparison

Note: Dashed lines depict the interval containing 68 percent of the individual forecasters pooled by Consensus.

Analogously, the forecasts for effective euro-area CPI inflation are shown in Figure 2.⁴ The forecasts of CPI inflation for 2008 were gradually re-estimated upwards. This was associated

³ Effective euro-area GDP growth (y-o-y) was 2.4% in 2007 and 1.0% in 2008.

⁴ Effective euro-area CPI inflation (y-o-y) reached 2.1% in 2007 and 2.8% in 2008.

with rising oil prices (Brent), which sharply increased in the first half of 2008 after passing the USD 90/b benchmark. An increase in the uncertainty of the mean Consensus forecast was observed in November and December 2007 (14 and 13 months before the end of the forecast horizon).

Finally, Figure 3 shows the evolution of the effective euro-area GDP growth and CPI inflation forecasts for 2009, when the financial crisis peaked. The forecasts for 2009 are characterized by the largest downward re-estimations, which are obvious given the impact of the crisis. In the case of the GDP growth forecast (see Figure 3A), the difference between the first published Consensus forecast (1.9 percent) and the last published Consensus forecast (-4.3 percent) reached 6.2 percentage points. The international institutions were slightly more pessimistic than Consensus.

The re-estimations were much smaller in the case of the CPI inflation forecast for 2009 (1.4 percentage points; see Figure 3B). The CPI inflation forecast was lifted initially due to increasing oil prices until July 2008, but afterwards it was lowered sharply as in the case of the GDP growth forecast (due to the consequences of the financial crisis).

In the case of both the GDP growth forecast and the CPI inflation forecast, the intervals around the mean Consensus forecast (the dashed lines) widened markedly after October 2008 (15 months before the end of the forecast horizon). This was associated with the increased uncertainty of the mean Consensus forecast at that time (increased disagreement among forecasters). A follow-up narrowing of the interval occurred in January 2009 in the case of CPI inflation (12 months before the end of the forecast horizon) and in June 2009 in the case of GDP growth (7 months before the end of the forecast horizon).



Figure 3: Effective Euro-Area GDP and CPI Forecasts – Visual Comparison

Note: Dashed lines depict the interval containing 68 percent of the individual forecasters pooled by Consensus.

Subsequently, Figures 4 and 5 below show a gradual improvement in the Consensus accuracy along the forecast horizon. The mean average percentage error (MAPE) serves as a measure of the forecasting accuracy, allowing us to compare the forecasts of different economic variables. Intuitively, the forecasting errors decrease as we move toward the end of the predicted year.

Forecasters have more information to make more accurate assumptions about the future and the forecasting process thus becomes easier. Similarly, Osterloh (2008) shows that the average RMSE of the Consensus forecasts for German GDP growth (across all forecasters and target years) diminishes strongly as we get closer to the end of the predicted year.

It is also apparent that Consensus is characterized by higher forecasting errors in the case of GDP growth than in the case of CPI inflation. The accuracy of the CPI inflation forecast is very similar in both periods (compare Figures 4 and 5). In contrast, the accuracy of the GDP growth forecast deteriorates substantially in the shorter period from 2004 to 2009. This is caused by the outlying year 2009 and confirms the stylized fact that some years are harder to predict than others.



5.2 Empirical Results

We tested for forecast bias in the Consensus forecasts, the international institutions' forecasts and the naïve forecasts (NF). Table 3 shows the results for the mean forecast error (MFE). The closer the value to zero, the less biased the forecast. Values in bold indicate the lowest forecasting error of the variables shown (CPI inflation and GDP growth) at the given forecast horizon (the current year and the next year).

Because the international institutions' forecasts are released twice a year only, the IMF forecasts are compared with the corresponding April and October Consensus issues and the EC, OECD and naïve forecasts are compared with the May and November Consensus issues (see also Section 4 for clarification).

		1994–2009				2002–2009			
MFE	Effecti	ve CPI	Effect	ive GDP	Effecti	ve CPI	Effective GDP		
	t	<i>t</i> +1	t	<i>t</i> +1	t	t+1	t	t+1	
CF (April, October)	0.02	-0.12	-0.07	-0.89***	0.08	0.06	-0.10	-1.11*	
IMF	0.04	0.06	-0.02	-1.03***	0.09	0.21	0.02	-1.17**	
CF (May, November)	0.01	-0.10	-0.07	-0.85***	0.07	0.07	-0.10	-1.04*	
EC	-	-	-	-	0.00	0.03	-0.06	-1.09**	
OECD	-	-	-0.03	-0.88***	0.07	0.22	0.00	-1.07**	
NF	-0.22	-0.26	-0.19	-0.32	-0.25	-0.16	-0.67	-0.93	

Table 3: Comparison of Forecasts: MFE

Note: - *t* is the forecast for the current year and t+1 is the forecast for the next year.

- MFE (mean forecast error) indicates whether a forecast is systematically biased. A positive value indicates that forecasts are, on average, underestimated.

- Equation (8) is estimated by OLS. Symbols ***, ** and * indicate rejection of the null hypothesis that the MFE is equal to zero at the 1%, 5% and 10% significance levels, respectively.

The forecasts for GDP growth are biased upwards (overestimated). This bias is statistically significant for the next-year forecasts with the exception of the naïve forecasts, which are characterized by the lowest mean forecasting errors. This is in accordance with the results found in the literature (Ager et al., 2009, Batchelor, 2001, Osterloh, 2008) and reflects the systematically positive expectations of forecasters about the future. Most Consensus forecasters are affiliated with investment banks and may intend to promote positive expectations among their clients (self-fulfilling expectations).

On the contrary, the CPI inflation forecasts are mostly underestimated by all institutions, including Consensus, but not at statistically significant levels, i.e., they are not biased. This holds especially for the shorter period from 2002 to 2009. This may possibly reflect a positive role of independent central banks and the resulting anchoring of inflation at low levels (Dovern et al., 2009).

		1994–2009 2002					-2009	
MAFE	Effecti	Effective CPI		ve GDP	Effecti	ve CPI	Effectiv	ve GDP
	t	t+1	t	t+1	t	t+1	t	t+1
CF (April, October)	0.18***	0.61***	0.4***	1.29***	0.18***	0.54***	0.34***	1.69***
IMF	0.2***	0.61***	0.42***	1.36***	0.21***	0.55***	0.38***	1.69***
CF (May, November)	0.14***	0.58***	0.34***	1.22***	0.15***	0.51***	0.29***	1.6***
EC	-	-	-	-	0.17***	0.5***	0.29***	1.64***
OECD	-	-	0.36***	1.19***	0.2***	0.62***	0.34***	1.47***
NF	0.66***	0.92***	1.42***	1.67***	0.68***	0.70***	1.47***	2.09***

Table 4: Comparison of Forecasts: MAFE

Note: - *t* is the forecast for the current year and t+1 is the forecast for the next year.

- MAFE (mean absolute forecast error) indicates the average size of the forecast errors (deviations) in the examined period irrespective of the direction (positive or negative) of the error.

- Equation (8) is estimated by OLS. Symbols ***, ** and * indicate rejection of the null hypothesis that the MAFE is equal to zero at the 1%, 5% and 10% significance levels, respectively.

It is evident from Table 4 that all the compared forecasts are biased at the 1% significance level if we take into account the absolute forecasting error (MAFE). Nevertheless, regarding solely the size of the forecasting errors, the Consensus forecasts are superior to the other forecasts in most cases. Finally, the naïve forecasts (NF) are characterized by having the largest forecasting errors, in accordance with our intuition.

	1994–2009				2002–2009			
RMSE	Effective CPI		Effectiv	ve GDP	Effect	ive CPI	Effectiv	ve GDP
	t	<i>t</i> +1	t	<i>t</i> +1	t	t+1	t	t+1
CF (April, October)	0.24***	0.76***	0.58**	1.84**	0.23**	0.75**	0.44***	2.32**
IMF	0.27***	0.74***	0.59***	1.84***	0.25**	0.7***	0.48**	2.22**
CF (May, November)	0.2***	0.72***	0.53**	1.75**	0.21**	0.7**	0.4**	2.2*
EC	-	-	-	-	0.22**	0.7**	0.39**	2.24*
OECD	-	-	0.50***	1.66**	0.26**	0.77**	0.44**	2.05*
NF	0.86***	1.08***	2.03***	2.34**	0.95*	0.87***	2.19*	2.84*

Table 5: Comparison of Forecasts: RMSE

Note: - *t* is the forecast for the current year and t+1 is the forecast for the next year.

- RMSE (root mean squared error) penalizes larger forecast errors more.

- Equation (8) is estimated by OLS. Symbols ***, ** and * indicate rejection of the null hypothesis that the MSE is equal to zero at the 1%, 5% and 10% significance levels, respectively.

In Table 5 we report RMSE statistics, which are expressed in annual percentage terms. The RMSE ranges from around 0.2 percentage points in the case of the Consensus CPI inflation forecast to 2.8 percentage points in the case of the naïve GDP growth forecast. As with the MAFE, all the forecasts are biased at statistically significant levels, but, again, the Consensus forecasting errors are relatively lower in many cases and, conversely, the naïve forecasts are the worst.

		1994-	-2009			2002–2009			
CF vs.	Effective CPI		Effectiv	ve GDP	Effecti	ve CPI	Effecti	ve GDP	
_	t	<i>t</i> +1	t	<i>t</i> +1	t	t+1	t	<i>t</i> +1	
EC	-	-	-	-	0.95	1.0	1.03	0.98***	
IMF	0.89	1.03	0.98	1.0	0.92	1.07	0.92	1.05	
OECD	-	-	1.06	1.05	0.81	0.91	0.91	1.07	
NF	0.23***	0.67***	0.26**	0.75	0.22	0.8^{***}	0.18	0.77	

Table 6: Relative RMSE and D-M Test of Statistical Significance in Forecast Differences

Note: - *t* is the forecast for the current year and t+1 is the forecast for the next year.

- A relative RMSE value lower than 1 indicates that Consensus performs better than the alternatives (EC, IMF, OECD, and NF), whereas a value higher than 1 indicates that the alternatives are better. Stars indicate if the null hypothesis of the same forecasting accuracy of the compared forecasts can be rejected at these levels of significance: ***1%, **5%, *10%.

- The IMF forecasts are compared with the April and October Consensus issues. The EC, OECD and NF are compared with the May and November Consensus issues.

The Consensus forecasts beat the international institutions' forecasts mainly in the current-year forecasts and, additionally, Consensus is superior to the naïve forecasts in all cases. Furthermore, if we test for the null hypothesis as to whether differences between the forecast accuracy of two competitive forecasts are on average the same, taking into account the number of observations and the volatility of variables (the D-M test; see Section 4), we see that the effective CPI inflation Consensus forecasts beat the naïve forecasts at the 1% significance level for both the current year and the next year (1994–2009). In the case of the GDP growth forecasts, Consensus beat the naïve forecast at the 5% significance level, but for the current-year forecast only. This finding is in accordance with the previous literature (Osterloh, 2008), which concludes that the accuracy of the Consensus next-year GDP growth forecasts is relatively low.

We are not able to make any strong conclusion about the differences between the Consensus forecasts on the one hand and the IMF and OECD forecasts on the other since they are not statistically significant. Assuming only the range of the RMSE forecasting errors, Consensus is more precise than the IMF forecasts for the current year but is less accurate than the OECD forecasts for GDP growth.

Finally, all the forecasts are covered in the shorter sample (2002–2009). Consensus beats the naïve forecast at the 1% significance level only in the case of the CPI inflation forecast for the next year. Nevertheless, Consensus is also superior (at the 1% significance level) to the EC forecast for GDP growth in the next year. In all remaining cases, even though the values of mostly lower than 1 in Table 6 point to Consensus having lower forecasting errors, this finding is not statistically significant.

If we focus solely on the pre-crisis period from 1994 to 2007 (see Annex 1, Table 4), the results are more diverse. Most importantly, Consensus not only beats, at a statistically significant level, the EC GDP growth forecasts, but also beats the IMF GDP growth forecast for both the current and the next year. Conversely, the Consensus forecast for GDP growth for the next year is outperformed (at the 10% significance level) by the OECD forecast.

		1994-	-2009			2002–2009			
CF vs.	Effecti	Effective CPI		ve GDP	Effecti	ive CPI	Effectiv	ve GDP	
-	t	<i>t</i> +1	t	<i>t</i> +1	t	<i>t</i> +1	t	t+1	
EC					0.75	0.67	0.47	6.8	
EC	-	-	-	-	(0.52)	(1.24)	(0.7)	(4.24)	
D.C	0.95**	1.01***	0.58	-0.2	0.77	-0.03	0.82^{*}	-2.9	
IIVIF	(0.35)	(0.36)	(0.42)	(2.08)	(0.46)	(0.7)	(0.4)	(2.09)	
OECD			0.37	-2.05*	0.95**	0.79	0.67**	-3.93**	
UECD	-	-	(0.29)	(1.19)	(0.33)	(0.56)	(0.27)	(1.7)	
NE	1.04***	0.71***	1.09***	1.06***	1.0***	1.33**	1.04***	1.5***	
NF	(0.04)	(0.16)	(0.05)	(0.22)	(0.06)	(0.46)	(0.05)	(0.43)	

Table 7: Test of the Information Content of Consensus Forecasts

Note: - *t* is the forecast for the current year and t+1 is the forecast for the next year.

- Equation (9) is estimated by OLS. If the presented coefficient is 1 or higher, then the alternative forecasts (EC, IMF, OECD, NF) add nothing to a combined forecast of Consensus and one of the alternative forecasts. The lower the coefficient, the less the amount of information contained in the Consensus forecasts. Standard errors are given in parenthesis below the coefficient estimates. Stars indicate rejection of the null hypothesis at these levels of significance: ***1%, **5%, *10%.

- The IMF forecasts are compared with the April and October Consensus issues. The EC, OECD and NF are compared with the May and November Consensus issues.

Additionally, we show the results of the information content test (Table 7). We are, *de facto*, carrying out a forecast comparison regression (see Equation 9, Section 4). If the presented coefficient is 1 or higher, then Consensus outperforms the alternative forecast (EC, IMF, OECD, and NF), which, in this particular case, adds nothing to a combined forecast.

The test reveals that Consensus outperforms the naïve forecasts at the 1% significance level in all cases. In the entire sample period (1994–2009), Consensus is also superior (at statistically significant levels) to the IMF forecasts for CPI inflation in both the current and the next year. Nevertheless, it is less useful than the OECD forecast for GDP growth in the next year at the 10% significance level.

In the shorter period (2002–2009), Consensus adds more information to a combined forecast with the IMF for GDP growth in the current year at the 10% significance level and in a combined forecast with the OECD for CPI inflation and GDP growth in the current year at the 5% significance level. Conversely, Consensus adds no information, or adds information of a perverse kind, to a combined forecast with the OECD for GDP growth in the next year at the 5% significance level.



Because the international institutions (EC, IMF and OECD) do not provide forecasts for PPI inflation, this is assessed only against the naïve forecast. Figures 6 and 7 show the forecast accuracy (measured by the mean percentage forecast error) of the April and October Consensus issues in the period from 2004 to 2009.⁵ The presented variables include effective euro-area CPI and PPI inflation and GDP growth.

The Consensus April and October forecasts for the current year are characterized by lower forecasting errors (MAPE) in comparison with the naïve forecasts (see Figure 6). The Consensus effective euro-area PPI inflation forecast for the current year is superior to the naïve forecast at the 10% significance level (measured by the D-M test statistic). The forecasting errors of the next-year forecasts (Figure 7) match the naïve forecasts except for GDP growth, where Consensus has a lower forecasting error.

If we compare the magnitude of the forecasting errors between different variables, we find that the Consensus forecasts for GDP growth and PPI inflation for the current year are more than four times higher compared to the CPI inflation forecast. The next-year forecasting errors of GDP growth and PPI inflation are more than six and three times higher, respectively, than the CPI inflation forecasts. This reflects the different historical variability of the two classes of economic variables.

If we observe the pre-crisis period only (see Annex 2), the differences in forecasting accuracy among individual variables are not as large. Furthermore, regarding only the size of the forecasting errors, the forecast for PPI inflation for the next year outperforms the naïve forecast.

5.3 Combined Forecast Accuracy

In this section we implement the method proposed by Eisenbeis, Waggoner and Zha (2002). This method compares forecasts which contain multiple variables. In our case, we have complete forecasts for two variables: effective euro-area CPI inflation and effective euro-area GDP growth. The method is based on the fact that some variables are hard to forecast due to their high

⁵ The short sample period is due to the limited availability of previous PPI forecasts.

historical volatility (e.g., GDP growth), whereas other variables, like CPI inflation, which has been well anchored in recent decades, are more easily predicted. When comparing different forecasts which contain outlooks for multiple variables, it may be useful to capture their historical volatility before carrying out a joint forecast assessment of them.

DMCE	1994-	-2009	2002–2009		
RWSE	t	<i>t</i> +1	t	<i>t</i> +1	
CF (April, October)	0.23***	0.79***	0.22****	0.86*	
IMF	0.25***	0.73***	0.23**	0.75**	
CF (May, November)	0.2***	0.75***	0.2**	0.8*	
EC	-	-	0.22****	0.82^{*}	
OECD	-	-	0.24**	0.8*	
NF	0.88^{**}	1.04***	1.06*	1.05**	

 Table 8: Comparison of Forecasts: Weighted Indicator (CPI, GDP)
 Particular

Note: - *t* is the forecast for the current year and t+1 is the forecast for the next year.

- Weighted Indicator= w_1 .CPI + w_2 .GDP. Weights w_1 and w_2 reflect historical volatility (statistical dispersion) in the period from 1994 to 2009 ($w_1 = 0.87$, $w_2 = 0.13$).

- Equation (8) is estimated by OLS. Symbols ***, ** and * indicate rejection of the null hypothesis that the MSE is equal to zero at the 1%, 5% and 10% significance levels, respectively.

In our case, the volatility (dispersion) of the actual values of effective euro-area GDP growth in the period from 1994 to 2009 was more than six-times higher⁶ than the volatility of effective euro-area CPI inflation. Obviously, due to its lower volatility, the forecasts of effective CPI inflation are expected to have lower forecasting errors. This is why we attach a six-times higher weighting to the forecasting errors of CPI inflation forecasts than to the GDP growth forecast. In fact, we penalize the CPI inflation forecast. After the weighting procedure we obtain a single indicator of both CPI and GDP forecasts.

Similarly to the analysis of sole variables, neither forecast significantly outperforms Consensus. Conversely, Consensus outperforms the EC forecast for the current year at the 5% significance level. Moreover, Consensus beats the naïve forecast at the 1% significance level in two cases (see Table 9).

⁶ The high difference in the volatility observed between the two variables is caused by the outlying year 2009. Until 2008, the historical volatility of GDP growth was only two times higher than the volatility of CPI inflation.

CE vc	1994-	-2009	2002–2009		
CF VS.	t	t+1	t	<i>t</i> +1	
EC	-	-	0.91**	0.98	
IMF	0.92	1.08	0.96	1.15	
OECD	-	-	0.83	1.0	
NF	0.23**	0.72	0.19	0.76***	

Table 9: Relative RMSE and D-M Test of	Statistical Significance in Forecast Differences:
Weighted Indicator (CPI, GDP)	

Note: - t is the forecast for the current year and t+1 is the forecast for the next year.

- A value of relative RMSE lower than 1 indicates that Consensus performs better than the alternatives (EC, IMF, OECD, and NF), whereas a value higher than 1 indicates that the alternatives are better. Stars indicate if the null hypothesis of the same forecast accuracy of the compared forecasts can be rejected at these levels of significance: ***1%, **5%, *10%.

- The IMF forecasts are compared with the April and October Consensus issues. The EC, OECD and NF are compared with the May and November Consensus issues.

If we exclude the crisis years of 2008 and 2009, the difference between the Consensus and international institutions' forecasts is less obvious (see Annex 3, Table 2). None of the alternative forecasts is significantly better or worse than Consensus. Nevertheless, Consensus is superior to the naïve forecast at the 1% significance level in all four cases.

Additionally, if we test for the information content of the alternative forecasts (Table 10), we find that Consensus is more informative than the naïve forecasts at the 1% significance level. Consensus adds more to a combined forecast in the case of the IMF (1994–2009), EC and OECD (2002–2009) current-year forecasts as well. Excluding the crisis years changes the overall picture only marginally.

CE vo	1994-	-2009	2002-2	2002–2009		
CF VS.	t	<i>t</i> +1	t	<i>t</i> +1		
EC			1.05*	1.23		
EC	-	-	(0.55)	(1.58)		
ME	0.83**	-0.38	0.67	-1.26		
llVIF	(0.35)	(0.6)	(0.45)	(0.79)		
OFCD			0.95**	0.47		
UECD	-	-	(0.33)	(0.71)		
NE	1.02***	1.58***	0.99***	2.06***		
N F	(0.04)	(0.26)	(0.05)	(0.47)		

 Table 10: Test of Information Content of Consensus Forecasts: Weighted Indicator (CPI, GDP)

Note: - *t* is the forecast is the forecast for the current year and t+1 is the forecast for the next year.

Equation (9) is estimated by OLS. If the presented coefficient is 1 or higher, then the alternative forecasts (EC, IMF, OECD, NF) add nothing to a combined forecast of Consensus and one of the alternative forecasts. The lower the coefficient, the less the amount of information contained in the Consensus forecasts. Standard errors are given in parenthesis below the coefficient estimates. Stars indicate rejection of the null hypothesis at these levels of significance: ***1%, **5%, *10%.

- The IMF forecasts are compared with the April and October Consensus issues. The EC, OECD and NF are compared with the May and November Consensus issues.

Another important characteristic of a multiple-economic-variable forecast is the expected consistency among single economic variables given by their long-term relationship. We measure the mutual consistency between the CPI inflation forecast and the GDP growth forecast by their mutual correlation. The correlation coefficient between actual values of CPI inflation and GDP growth in the period from 1994 to 2009 was positive (corr. coef. equal to 0.51). Nevertheless, the correlation coefficient was strongly affected by the last observation. If we exclude 2009, no correlation between these two variables is apparent.

Accordingly, the Consensus and international institutions' forecasts for the current year are correlated at similar or higher levels to the historical values. They are therefore strongly affected by the actual values. Conversely, the forecasts for the next year are less correlated. Thus, we do not detect strong consistency in the longer-term forecasts.

Furthermore, if we look at the direction of the forecast re-estimations we cannot find strong synchronization between the forecast updates of CPI inflation and GDP growth. The forecasts for these two variables are very often re-estimated in different directions. Only about 54 percent of all the Consensus re-estimations are synchronized, i.e., the forecasts for CPI inflation and GDP growth are re-estimated in the same direction. The degree of synchronization in the case of the EC, IMF and OECD forecasts varied between 39 and 56 percent, depending on the number of observations.

6. USD/EUR Forecast Accuracy

Figure 8 shows the evolution of the nominal USD/EUR exchange rate and the corresponding Consensus forecasts.⁷ The Consensus point forecasts at the three-month, one-year and two-year horizons are interpolated. We start with the January 1999 Consensus forecast, which extends from April 1999 to January 2001. All subsequent Consensus forecasts are constructed in the same way.

Figure 8: USD/EUR and its Consensus Forecasts (1999–2009)



Source: Thomson's Datastream and Consensus Economics

At the beginning of the observed period, when the US dollar was appreciating, the Consensus forecasts were systematically higher, i.e., a weaker dollar was systematically predicted. As from 2002, when the trend reversed, the dollar was depreciating faster than Consensus assumed. When the dollar moved above its average value for the whole period (USD/EUR 1.18), Consensus started to forecast stability or appreciation of the dollar. Finally, after the dollar reached historical lows (in summer 2008) Consensus started to systematically predict its appreciation.

Restricted to our sample period from 1999 to 2009, the Consensus forecasts tended systematically towards the average USD/EUR value. Accordingly, the disparity between the one-year and two-years-ahead forecasts was negligible. They both reached USD/EUR 1.2 on average.

⁷ Only the January, April, July and October Consensus forecasts (out of a total of twelve monthly forecasts) are presented to avoid overloading the figure.

			1999–2009) ^{a)}	2002–2009 ^{a)}			
Forecast	horizon	<i>3M</i>	1Y	2Y	3М	1Y	2Y	
	CF	-0.002	0.006	0.051***	0.019**	0.063***	0.118***	
MFE	FWD	0.008	0.032***	0.077^{***}	0.022***	0.068^{***}	0.114***	
	NF	0.008	0.036***	0.095***	0.021**	0.069***	0.125***	
	CF	0.066****	0.116***	0.146***	0.066***	0.101***	0.125***	
MAFE	FWD	0.06***	0.119***	0.152***	0.063***	0.123***	0.135***	
	NF	0.062***	0.118***	0.153***	0.068^{***}	0.127***	0.145***	
	CF	0.081***	0.138***	0.177***	0.081***	0.122***	0.157***	
RMSE	FWD	0.075***	0.137***	0.188***	0.079***	0.141***	0.177^{***}	
	NF	0.079***	0.135***	0.187***	0.085***	0.144***	0.186***	

Table 11: Comparison of Forecasts: MFE, MAFE, RMSE

Note: - ^{a)} 3M (3 months) ahead forecasts are assessed until the December 2009 forecast, 1Y (1 year) ahead forecasts are assessed until the March 2009 forecast and 2Y (2 years) ahead forecasts are assessed until the March 2008 forecast.

- CF: Consensus forecast, FWD: forecast derived from forward exchange rates on the survey day of the Consensus forecast, NF: naïve forecast.

- Equation (8) is estimated by OLS. Symbols ***, ** and * indicate rejection of the null hypothesis that the MSE is equal to zero at the 1%, 5% and 10% significance levels, respectively.

From Table 11 (MFE), it is apparent that all forecasts were unbiased only in the case of the threemonths-ahead forecasts in the long sample from 1999 to 2009. In addition, the one-year-ahead Consensus forecast was unbiased as well. The remaining forecasts were biased downwards, i.e., a stronger dollar was systematically predicted.

The Consensus forecasting accuracy was compared with the naïve forecast and the forecast derived from the forward exchange rate (FWD). Both forecasts were even more biased, except for the FWD forecast at the two-year horizon in the short sample period (2002–2009).

CF vs.		1999–2009 ^{a)}			$2002-2009^{a}$			
	3M	1Y	2Y	3M	1Y	2Y		
FWD	1.08^{**}	1.01	0.94	1.03	0.87^{**}	0.89		
NF	1.03	1.02	0.95	0.95^{*}	0.85***	0.84^{*}		

Table 12: Relative	RMSE and D-M	Test of Statistica	l Significance in	Forecast Differences

Note: - ^{a)} 3M (3 months) ahead forecasts are assessed until the December 2009 forecast, 1Y (1 year) ahead forecasts are assessed until the March 2009 forecast and 2Y (2 years) ahead forecasts are assessed until the March 2008 forecast.

- CF: Consensus forecast, FWD: forecast derived from forward exchange rates on the survey day of the Consensus forecast, NF: naïve forecast.

- A value of relative RMSE lower than 1 indicates that Consensus performs better than the alternatives (EC, IMF, OECD, and NF), whereas a value higher than 1 indicates that the alternatives are better. Stars indicate if the null hypothesis of the same forecast accuracy of the compared forecasts can be rejected at these levels of significance: ***1%, **5%, *10%.

Looking at Table 12, the Consensus three-months-ahead forecast is outperformed by the FWD forecast at the 5% significance level in the whole sample (1999–2009). Nevertheless, in the short sample period from 2002 to 2009, the Consensus forecast improved relatively to both the FWD and the naïve forecast. Consensus was, in particular, superior to the naïve forecast at all three forecast horizons and, in addition, against the FWD forecast at the one-year horizon.

If we exclude the crisis years (see Annex 4, Table 2), the results are the same in the long sample period from 1999 to 2007. Nevertheless, the results for the short sample period from 2002 to 2007 confirm the statistically significant superiority of the Consensus forecast (in accordance with Table 12).

CE vs		1999–2009 ^{a)}		2002–2009 ^{a)}			
Cr vs.	3M	1Y	2Y	3M	1Y	2Y	
FWD	-0.22	0.44*	0.94***	0.19	1.01***	2.04***	
	(0.23)	(0.22)	(0.25)	(0.37)	(0.06)	(0.37)	
NF	0.22	0.41**	0.79***	1.44***	1.87***	2.45***	
	(0.28)	(0.19)	(0.2)	(0.43)	(0.27)	(0.31)	

 Table 13: Test of Information Content of Consensus Forecasts

Note: Equation (9) is estimated by OLS. If the presented coefficient is 1 or higher, then the alternative forecasts (FWD and NF) add nothing to a combined forecast of Consensus and one of the alternative forecasts. The lower the coefficient, the less the amount of information contained in the Consensus forecasts. Standard errors are given in parenthesis below the coefficient estimates. Stars indicate rejection of the null hypothesis at these levels of significance: ***1%, **5%, *10%.

Furthermore, the results of the information content test (Table 13) support the superiority of Consensus. In the short sample period Consensus is superior to the alternative forecasts at the 1% significance level at all forecast horizons except for the FWD three-months-ahead forecast. In the long sample period, Consensus outperforms the alternative forecasts in the two-years-ahead forecast at the 1% significance level.

7. Conclusion

We compared the accuracy of the Consensus Economics forecasts to those of the European Commission, International Monetary Fund and Organization for Economic Co-operation and Development, and also to the naïve forecast and the forecast implied by forward exchange rates. Basic descriptive statistics of forecast efficiency and forecast bias as well as tests for statistical significance in the differences between competing forecasts were applied. We analyzed the forecasts for GDP growth, CPI inflation, PPI inflation and USD/EUR.

Forecasting accuracy was compared before and during the recent financial crisis. With respect to the GDP growth and CPI inflation forecasts, we found that the Consensus Economics forecasts usually outperformed the others significantly, especially during the whole period from 1994 to 2009. The results were not that strong for the pre-crisis period. The Consensus forecasts beat the international institutions' forecasts mainly in the current-year forecasts and, additionally, Consensus is superior to the naïve forecasts at all forecast horizons.

In accordance with previous literature, we confirm a relatively low level of accuracy of the nextyear GDP growth forecasts, which are biased upwards by all institutions, whereas the CPI inflation forecasts are unbiased.

The PPI inflation and USD/EUR forecasts were compared only with the naïve forecast and the forward rate forecast in the case of USD/EUR due to the absence of comparable forecasts by international institutions. The Consensus PPI inflation forecast significantly beats the naïve forecast for the current year. The Consensus forecast for USD/EUR improved dramatically after 2002, when it significantly outperformed the naïve forecast at all forecasting horizons and the forward implied forecast at the one-year horizon. A stronger dollar was systematically predicted over the observed period by all the assessed forecasts.

The practical advantage of the Consensus Economics forecasts lies in a broader range of predicted economic variables and in the high frequency of forecast releases (every month). Based on the relatively high forecasting accuracy and the additional practical characteristics of Consensus, the Czech National Bank will continue to use it in its prediction process. However, the Consensus Forecasts will also be regularly confronted with alternative assumptions.

References

- AGER, P., M. KAPPLER, AND S. OSTERLOH (2009): "The Accuracy and Efficiency of the Consensus Forecasts: A Further Application and Extension of the Pooled Approach." *International Journal of Forecasting* Vol. 25, Iss. 1, pp. 167–181.
- ANG, A., G. BEKAERT, AND M. WEI (2007): "Do Macro Variables, Asset Markets, or Surveys Forecast Inflation Better?" *Journal of Monetary Economics* Vol. 54, pp. 1163–1212.
- ANTAL, J., M. HLAVÁČEK, AND R. HORVÁTH (2008): "Do Central Bank Forecast Errors Contribute to the Missing of Inflation Targets? The Case of the Czech Republic." *Czech Journal of Economics and Finance* Vol. 58, Iss. 9–10, pp. 434–453.
- BABECKÝ, J. AND J. PODPIERA (2011): "Inflation Forecast Errors in the Czech Republic: Evidence from a Panel of Institutions." *Eastern European Economics*, forthcoming.
- BATCHELOR R. (2001): "How Useful are the Forecasts of Intergovernmental Agencies? The IMF and OECD Versus the Consensus." *Applied Economics* Vol. 33, Iss. 2, pp. 225–235.
- BATCHELOR, R. (2007): "Bias in Macroeconomic Forecasts." *International Journal of Forecasting* Vol. 23, Iss. 2, pp. 189–203.
- BOWLES, C., R. FRIZ, V. GENRE, G. KENNY, A. MEYLER, AND T. RAUTANEN (2007): "The ECB Survey of Professional Forecasters – A Review after Eight Years' Experience." ECB Occasional Paper Series No. 59.
- CROUSHORE, D. (2010): "An Evaluation of Inflation Forecasts from Surveys Using Real-Time Data." *The B.E. Journal of Macroeconomics* Vol. 10, Iss. 1, pp. 1–30.
- DIEBOLD, F. X. AND R.S. MARIANO (1995): "Comparing Predictive Accuracy." Journal of Business & Economic Statistics Vol. 13, No. 3, pp. 253–263.
- DOVERN, J., U. FRITSCHE, AND J. SLACALEK (2009): "Disagreement among Forecasters in G7 Countries." ECB Working Paper No. 1082.
- EISENBEIS, R., D. WAGGONER, AND T. ZHA (2002): "Evaluating Wall Street Journal Forecasters: A Multivariate Approach." WP 2002–8a. Federal Reserve Bank of Atlanta.
- OSTERLOH, S. (2008): "Accuracy and Properties of German Business Cycle Forecasts." *Applied Economics Quarterly* Vol. 54, No. 1, pp. 27–57.
- STOCK, J. AND M. WATSON (1999): "Forecasting Inflation." *Journal of Monetary Economics* Vol. 44, Iss. 2, pp. 293–335.
- TIMMERMANN, A. (2006): "An Evaluation of the World Economic Outlook Forecasts." IMF Working Paper 06/59.

Annex 1: MFE, MAFE, RMSE and D-M Test of Statistical Significance (1994–2007)

	1994–2007				2002–2007			
MFE	Effecti	Effective CPI		Effective GDP		ive CPI	Effective GDP	
	t	t+1	t	<i>t</i> +1	t	<i>t</i> +1	t	<i>t</i> +1
CF (April, October)	0.02	-0.1	-0.02	-0.52**	0.11	0.16	0.01	-0.38
IMF	0.04	0.08	-0.01	-0.72***	0.1	0.3**	0.06	-0.54
CF (May, November)	0.02	-0.08	-0.04	-0.49**	0.09	0.16	-0.02	-0.33
EC	-	-	-	-	0.04	0.14	-0.01	-0.37
OECD	-	-	-0.03	-0.57***	0.13*	0.39**	0.01	-0.46
NF	-0.13	-0.24	0.28	0.29	-0.05	-0.1	0.27	0.21

Table 1: Comparison of Forecasts: MFE (corresponds to Table 3, Section 4.2)

Note: - *t* is the forecast for the current year and t+1 is the forecast for the next year.

- MFE (mean forecast error) indicates whether a forecast is systematically biased. A positive value indicates that forecasts are on average underestimated.

- Equation (8) is estimated by OLS. Symbols ***, ** and * indicate rejection of the null hypothesis that the MFE is equal to zero at the 1%, 5% and 10% significance levels, respectively.

Table 2: Comparison of Forecasts: MAFE (corresponds to Table 4, Section 4.2)

	1994–2007				2002-2007				
MAFE	Effective CPI		Effectiv	Effective GDP		Effective CPI		Effective GDP	
	t	t+1	t	<i>t</i> +1	t	<i>t</i> +1	t	<i>t</i> +1	
CF (April, October)	0.18***	0.5***	0.39***	0.98***	0.19***	0.28***	0.31***	1.15***	
IMF	0.2^{***}	0.54***	0.43***	1.1^{***}	0.2***	0.37***	0.38***	1.23***	
CF (May, November)	0.14***	0.48***	0.35***	0.93***	0.16***	0.26***	0.28***	1.08^{***}	
EC	-	-	-	-	0.17^{***}	0.25***	0.25***	1.1^{***}	
OECD	-	-	0.34***	0.93***	0.2^{***}	0.49***	0.28^{***}	1.0***	
NF	0.54^{***}	0.85***	1.12***	1.26***	0.41***	0.47^{***}	0.8^{***}	1.34***	

Note: - *t* is the forecast for the current year and t+1 is the forecast for the next year.

- MAFE (mean absolute forecast error) indicates the average size of the forecast errors (deviations) in the examined period irrespective of the direction (positive or negative) of the error.

- Equation (8) is estimated by OLS. Symbols ***, ** and * indicate rejection of the null hypothesis that the MAFE is equal to zero at the 1%, 5% and 10% significance levels, respectively.

Table 3: Comparison of Forecasts: RMSE (corresponds to Table 5, Section 4.2)

	1994–2007				2002–2007			
RMSE	Effective CPI		Effective GDP		Effective CPI		Effective GDP	
	t	<i>t</i> +1	t	<i>t</i> +1	t	t+1	t	t+1
CF (April, October)	0.25***	0.62***	0.6**	1.2***	0.25**	0.36**	0.42**	1.34***
IMF	0.27^{***}	0.67^{***}	0.61***	1.34***	0.26^{*}	0.5^{**}	0.51^{*}	1.43***
CF (May, November)	0.21***	0.59***	0.54**	1.15***	0.23**	0.34**	0.4^{*}	1.27**
EC	-	-	-	-	0.23**	0.35**	0.37^{*}	1.31**
OECD	-	-	0.48***	1.13***	0.28^{**}	0.62^{**}	0.37**	1.21**
NF	0.65***	1.01***	1.54**	1.54***	0.49**	0.58^{**}	0.99**	1.48***

Note: - *t* is the forecast for the current year and t+1 is the forecast for the next year.

- RMSE (root mean squared error) penalizes larger forecast errors more.

- Equation (8) is estimated by OLS. Symbols ***, ** and * indicate rejection of the null hypothesis that the MSE is equal to zero at the 1%, 5% and 10% significance levels, respectively.

		1994	-2007			2002–2007				
CF vs.	Effectiv	Effective CPI		Effective GDP		Effective CPI		Effective GDP		
_	t	<i>t</i> +1	t	<i>t</i> +1	t	<i>t</i> +1	t	<i>t</i> +1		
EC	-	-	-	-	1.0	0.97	1.08	0.97^{**}		
IMF	0.93	0.93	0.98	0.9	0.96	0.72	0.82***	0.94**		
OECD	-	-	1.13	1.02	0.82	0.55	1.08	1.05^{*}		
NF	0.32***	0.58	0.35**	0.75	0.47**	0.59**	0.4	0.86		

Table 4: Relative RMSE and D-M Test of Statistical Significance in Forecast Differences (corresponds to Table 6, Section 4.2)

Note: - *t* is the forecast for the current year and t+1 is the forecast for the next year.

- A relative RMSE value lower than 1 indicates that Consensus performs better than the alternatives (EC, IMF, OECD, and NF), whereas a value higher than 1 indicates that the alternatives are better. Stars indicate if the null hypothesis of the same forecasting accuracy of the compared forecasts can be rejected at these levels of significance: ***1%, **5%, *10%.

- The IMF forecasts are compared with the April and October Consensus issues. The EC, OECD and NF are compared with the May and November Consensus issues.

		1994-	-2007		2002–2007			
CF vs.	Effecti	Effective CPI		Effective GDP		Effective CPI		ve GDP
	t	t+1	t	<i>t</i> +1	t	<i>t</i> +1	t	<i>t</i> +1
EC				-	0.41	0.68	-1.5	4.74
	-	-	-		(0.92)	(0.72)	(1.45)	(3.42)
ŊŒ	0.9**	1.01**	0.82	2.94***	0.63	1.51***	3.54***	2.98
IIVIF	(0.39)	(0.46)	(0.67)	(0.77)	(0.56)	(0.42)	(0.72)	(1.7)
OECD			-0.03	0.19	1.07^{**}	1.96***	0.32	-0.65
UECD	-	-	(0.38)	(0.97)	(0.48)	(0.21)	(0.42)	(1.45)
NF	1.14***	1.34***	1.16***	0.75***	1.11***	1.08***	1.05***	0.7^{**}
	(0.07)	(0.17)	(0.08)	(0.15)	(0.17)	(0.23)	(0.14)	(0.31)

Table 5: Test of Information Content of CF Forecasts

Note: - *t* is the forecast for the current year and t+1 is the forecast for the next year.

- Equation (9) is estimated by OLS. If the presented coefficient is 1 or higher, then the alternative forecasts (EC, IMF, OECD, NF) add nothing to a combined forecast of Consensus and one of the alternative forecasts. The lower the coefficient, the less the amount of information contained in the Consensus forecasts. Standard errors are given in parenthesis below the coefficient estimates. Stars indicate rejection of the null hypothesis at these levels of significance: ***1%, **5%, *10%.

- The IMF forecasts are compared with the April and October Consensus issues. The EC, OECD and NF are compared with the May and November Consensus issues.

Annex 2: MAPE: Comparison of Consensus Forecasts and Naïve Forecasts (2004 - 2007)



Annex 3: Combined Forecast Accuracy (1994–2007)

1 0	0			
DMSE	1994-	2002	2002-2007	
RWSE	t	<i>t</i> +1	t	<i>t</i> +1
CF (April, October)	0.26***	0.6***	0.23**	0.50***
IMF	0.26***	0.6***	0.22^{*}	0.46**
CF (May, November)	0.24***	0.57***	0.22**	0.46***
EC	-	-	0.21**	0.48***
OECD	-	-	0.22^{**}	0.52***
NF	0.6***	0.8^{***}	0.49**	0.76***

Table 1: Comparison of Forecasts: Weighted Indicator (CPI, GDP)

Note: - *t* is the forecast for the current year and t+1 is the forecast for the next year.

- Weighted Indicator=w₁.CPI + w₂.GDP. Weights w₁ and w₂ reflect historical volatility (statistical dispersion) in the period from 1994 to 2009 ($w_1 = 0.87, w_2 = 0.13$).

- Equation (8) is estimated by OLS. Symbols ***, ** and * indicate rejection of the null hypothesis that the MSE is equal to zero at the 1%, 5% and 10% significance levels, respectively.

CF vs	1994	-2007	2002–2007		
CF VS.	t	t+1	t	<i>t</i> +1	
EC	-	-	1.05	0.96	
IMF	1.0	1.0	1.05	1.09	
OECD	-	-	1.0	0.88	
NF	0.4***	0.71***	0.45***	0.61***	

 Table 2: Relative RMSE and D-M Test of Statistical Significance in Forecast Differences:

 Weighted Indicator (CPI, GDP)

Note: - *t* is the forecast for the current year and t+1 is the forecast for the next year.

- A relative RMSE value lower than 1 indicates that Consensus performs better than the alternatives (EC, IMF, OECD, and NF), whereas a value higher than 1 indicates that the alternatives are better. Stars indicate if the null hypothesis of the same forecast accuracy of the compared forecasts can be rejected at these levels of significance: ***1%, **5%, *10%.

- The IMF forecasts are compared with the April and October Consensus issues. The EC, OECD and NF are compared with the May and November Consensus issues.

	1994-	-2007	2002-	2002–2007		
CF VS.	t	t+1	t	<i>t</i> +1		
EC			0.29	0.79		
	-	-	(0.99)	(0.89)		
DAE	0.75^{*}	0.72	0.34	0.87		
IMF	(0.43)	(0.54)	(0.65)	(0.58)		
OECD			0.85^{*}	1.7***		
OECD	-	-	(0.46)	(0.35)		
NE	1.14***	1.39***	1.12***	1.48***		
INΓ	(0.08)	(0.2)	(0.17)	(0.23)		

 Table 3: Test of Information Content of Consensus Forecasts: Weighted Indicator (CPI, GDP)

Note: - *t* is the forecast for the current year and t+1 is the forecast for the next year.

Equation (9) is estimated by OLS. If the presented coefficient is 1 or higher, then the alternative forecasts (EC, IMF, OECD, NF) add nothing to a combined forecast of Consensus and one of the alternative forecasts. The lower the coefficient, the less the amount of information contained in the Consensus forecasts. Standard errors are given in parenthesis below the coefficient estimates. Stars indicate rejection of the null hypothesis at these levels of significance: ***1%, **5%, *10%.
 The IMF forecasts are compared with the April and October Consensus issues. The EC, OECD

and NF are compared with the May and November Consensus issues.

Annex 4: USD/EUR Forecast Accuracy (1999–2007)

		1999–2007 ^{a)}			$2002 - 2007^{a)}$			
Forecast Horizon		3М	1Y	2Y	<i>3M</i>	1Y	2Y	
	CF	-0.004	-0.009	0.032	0.024***	0.064***	0.124***	
MFE	FWD	0.01^{*}	0.033**	0.079***	0.031***	0.087^{***}	0.137***	
	NF	0.013**	0.042***	0.095***	0.033***	0.092***	0.144***	
	CF	0.058***	0.115***	0.154***	0.053***	0.092***	0.128***	
MAFE	FWD	0.052***	0.113***	0.165***	0.052***	0.114***	0.147***	
	NF	0.053***	0.109***	0.16***	0.055***	0.117***	0.152***	
	CF	0.07^{***}	0.136***	0.184***	0.063***	0.11***	0.159***	
RMSE	FWD	0.063***	0.131***	0.199***	0.063***	0.133***	0.191***	
	NF	0.064***	0.126***	0.191***	0.066***	0.134***	0.193***	

Table 1: Comparison of Forecasts: MFE, MAFE, RMSE

Note: - ^{a)} 3M (3 months) ahead forecasts are assessed until the December 2009 forecast, 1Y (1 year) ahead forecasts are assessed until the March 2009 forecast and 2Y (2 years) ahead forecasts are assessed until the March 2008 forecast.

- CF: Consensus forecast, FWD: forecast derived from forward exchange rates on the survey day of the Consensus forecast, NF: naïve forecast.

- Equation (8) is estimated by OLS. Symbols ***, ** and * indicate rejection of the null hypothesis that the MSE is equal to zero at the 1%, 5% and 10% significance levels, respectively.

Table 2: Relative RMSE and D-M Test of Statistical Significance in Forecast Differences

CF vs.	1999–2007 ^{a)}		2002–2007 ^{a)}			
	3M	1Y	2Y	3M	1Y	2Y
FWD	1.11	1.04	0.92	1.0	0.83**	0.83
NF	1.09	1.08	0.96	0.95**	0.82***	0.82^{*}

Note: - ^{a)} 3M (3 months) ahead forecasts are assessed until the December 2009 forecast, 1Y (1 year) ahead forecasts are assessed until the March 2009 forecast and 2Y (2 years) ahead forecasts are assessed until the March 2008 forecast.

- CF: Consensus forecast, FWD: forecast derived from forward exchange rates on the survey day of the Consensus forecast, NF: naïve forecast.

- A relative RMSE value lower than 1 indicates that Consensus performs better than the alternatives (EC, IMF, OECD, and NF), whereas a value higher than 1 indicates that the alternatives are better. Stars indicate if the null hypothesis of the same forecast accuracy of the compared forecasts can be rejected at these levels of significance: ***1%, **5%, *10%.

CEve	1999–2007 ^{a)}			2002–2007 ^{a)}		
Cr vs.	3M	1Y	2Y	3M	1Y	2Y
FWD	-0.24	0.28	1.06***	0.51	2.98***	3.8***
	(0.24)	(0.24)	(0.28)	(0.41)	(0.34)	(0.4)
NF	-0.22	0.18	0.68***	1.37***	2.68***	3.64***
	(0.27)	(0.2)	(0.23)	(0.5)	(0.32)	(0.37)

Table 3: Test of Information Content of Consensus Forecasts

Note: Equation (9) is estimated by OLS. If the presented coefficient is 1 or higher, then the alternative forecasts (FWD and NF) add nothing to a combined forecast of the Consensus forecast and one of the alternative forecasts. The lower the coefficient, the less the amount of information contained in the Consensus forecasts. Standard errors are given in parenthesis below the coefficient estimates. Stars indicate rejection of the null hypothesis at these levels of significance: ***1%, **5%, *10%.

CNB WORKING PAPER SERIES

14/2010	Filip Novotný Marie Raková	Assessment of consensus forecasts accuracy: The Czech National Bank perspective
13/2010	Jan Filáček Branislav Saxa	Central bank forecasts as a coordination device
12/2010	Kateřina Arnoštová David Havrlant Luboš Růžička Peter Tóth	Short-term forecasting of Czech quarterly GDP using monthly indicators
11/2010	Roman Horváth Kateřina Šmídková Jan Zápal	Central banks' voting records and future policy
10/2010	Alena Bičáková Zuzana Prelcová Renata Pašaličová	Who borrows and who may not repay?
9/2010	Luboš Komárek Jan Babecký Zlatuše Komárková	Financial integration at times of financial instability
8/2010	Kamil Dybczak Peter Tóth David Voňka	Effects of price shocks to consumer demand. Estimating the QUAIDS demand system on Czech Household Budget Survey data
7/2010	Jan Babecký Philip Du Caju Theodora Kosma Martina Lawless Julián Messina Tairi Rõõm	The margins of labour cost adjustment: Survey evidence from European Firms
6/2010	Tomáš Havránek Roman Horváth Jakub Matějů	Do financial variables help predict macroeconomic environment? The case of the Czech Republic
5/2010	Roman Horváth Luboš Komárek Filip Rozsypal	Does money help predict inflation? An empirical assessment for Central Europe
4/2010	Oxana Babecká Kucharčuková Jan Babecký Martin Raiser	A Gravity approach to modelling international trade in South- Eastern Europe and the Commonwealth of Independent States: The role of geography, policy and institutions
3/2010	Tomáš Havránek Zuzana Iršová	Which foreigners are worth wooing? A Meta-analysis of vertical spillovers from FDI
2/2010	Jaromír Baxa Roman Horváth Bořek Vašíček	How does monetary policy change? Evidence on inflation targeting countries
1/2010	Adam Geršl Petr Jakubík	Relationship lending in the Czech Republic
15/2009	David N. DeJong Roman Liesenfeld Guilherme V. Moura Jean-Francois Richard Hariharan Dharmarajan	Efficient likelihood evaluation of state-space representations

14/2009	Charles W. Calomiris	Banking crises and the rules of the game
13/2009	Jakub Seidler Petr Jakubík	The Merton approach to estimating loss given default: Application to the Czech Republic
12/2009	Michal Hlaváček Luboš Komárek	Housing price bubbles and their determinants in the Czech Republic and its regions
11/2009	Kamil Dybczak Kamil Galuščák	Changes in the Czech wage structure: Does immigration matter?
10/2009	Jiří Böhm Petr Král Branislav Saxa	Percepion is always right: The CNB's monetary policy in the media
9/2009	Alexis Derviz Marie Raková	Funding costs and loan pricing by multinational bank affiliates
8/2009	Roman Horváth Anca Maria Podpiera	Heterogeneity in bank pricing policies: The Czech evidence
7/2009	David Kocourek Filip Pertold	The impact of early retirement incentives on labour market participation: Evidence from a parametric change in the Czech Republic
6/2009	Nauro F. Campos Roman Horváth	Reform redux: Measurement, determinants and reversals
5/2009	Kamil Galuščák Mary Keeney Daphne Nicolitsas Frank Smets Pawel Strzelecki Matija Vodopiyec	The determination of wages of newly hired employees: Survey evidence on internal versus external factors
4/2009	Jan Babecký Philip Du Caju Theodora Kosma Martina Lawless Julián Messina Tairi Rõõm	Downward nominal and real wage rigidity: Survey evidence from European firms
3/2009	Jiri Podpiera Laurent Weill	Measuring excessive risk-taking in banking
2/2009	Michal Andrle Tibor Hlédik Ondra Kameník Jan Vlček	Implementing the new structural model of the Czech National Bank
1/2009	Kamil Dybczak Jan Babecký	The impact of population ageing on the Czech economy
14/2008	Gabriel Fagan Vitor Gaspar	Macroeconomic adjustment to monetary union
13/2008	Giuseppe Bertola Anna Lo Prete	Openness, financial markets, and policies: Cross-country and dynamic patterns
12/2008	Jan Babecký Kamil Dybczak Kamil Galuščák	Survey on wage and price formation of Czech firms
11/2008	Dana Hájková	The measurement of capital services in the Czech Republic
10/2008	Michal Franta	<i>Time aggregation bias in discrete time models of aggregate duration data</i>

9/2008	Petr Jakubík Christian Schmieder	Stress testing credit risk: Is the Czech Republic different from Germany?
8/2008	Sofia Bauducco Aleš Bulíř Martin Čihák	Monetary policy rules with financial instability
7/2008	Jan Brůha Jiří Podpiera	The origins of global imbalances
6/2008	Jiří Podpiera Marie Raková	The price effects of an emerging retail market
5/2008	Kamil Dybczak David Voňka Nico van der Windt	The effect of oil price shocks on the Czech economy
4/2008	Magdalena M. Borys Roman Horváth	The effects of monetary policy in the Czech Republic: An empirical study
3/2008	Martin Cincibuch Tomáš Holub Jaromír Hurník	Central bank losses and economic convergence
2/2008	Jiří Podpiera	Policy rate decisions and unbiased parameter estimation in conventionally estimated monetary policy rules
1/2008	Balázs Égert Doubravko Mihaljek	Determinants of house prices in Central and Eastern Europe
17/2007	Pedro Portugal	U.S. unemployment duration: Has long become longer or short become shorter?
16/2007	Yuliya Rychalovská	Welfare-based optimal monetary policy in a two-sector small open economy
15/2007	Juraj Antal František Brázdik	The effects of anticipated future change in the monetary policy regime
14/2007	Aleš Bulíř Kateřina Šmídková Viktor Kotlán David Navrátil	Inflation targeting and communication: Should the public read inflation reports or tea leaves?
13/2007	Martin Cinncibuch Martina Horníková	<i>Measuring the financial markets' perception of EMU enlargement:</i> <i>The role of ambiguity aversion</i>
12/2007	Oxana Babetskaia- Kukharchuk	Transmission of exchange rate shocks into domestic inflation: The case of the Czech Republic
11/2007	Jan Filáček	Why and how to assess inflation target fulfilment
10/2007	Michal Franta Branislav Saxa Kateřina Šmídková	Inflation persistence in new EU member states: Is it different than in the Euro area members?
9/2007	Kamil Galuščák Jan Pavel	Unemployment and inactivity traps in the Czech Republic: Incentive effects of policies
8/2007	Adam Geršl Ieva Rubene Tina Zumer	Foreign direct investment and productivity spillovers: Updated evidence from Central and Eastern Europe
7/2007	Ian Babetskii Luboš Komárek Zlatuše Komárková	Financial integration of stock markets among new EU member states and the euro area
6/2007	Anca Pruteanu-Podpiera	Market power and efficiency in the Czech banking sector

14/2005	Stephen G. Cecchetti	The brave new world of central banking: The policy challenges posed by asset price booms and busts
1/2006	Ian Babetskii	Aggregate wage flexibility in selected new EU member states
2/2006	Kamil Dybczak	Generational accounts in the Czech Republic
3/2006	Aleš Krejdl	Fiscal sustainability – definition, indicators and assessment of Czech public finance sustainability
	Vladislav Flek Dana Hájková Jaromír Hurník	(1995–2005)
4/2006	Kamil Dybczak	Supply-side performance and structure in the Czech Republic
5/2006	Martin Fukač	New Keynesian model dynamics under heterogeneous expectations
6/2006	Martin Fukač Adrian Pagan	Issues in adopting DSGE models for use in the policy process
7/2006	Alena Bičáková Jiří Slačálek Michal Slavík	Fiscal implications of personal tax adjustments in the Czech Republic
8/2006	Aleš Bulíř Jaromír Hurník	The Maastricht inflation criterion: "Saints" and "Sinners"
9/2006	Alexis Derviz Jiří Podpiera	Cross-border lending contagion in multinational banks
10/2006	Jiří Podpiera Marie Raková	Degree of competition and export-production relative prices when the exchange rate changes: Evidence from a panel of Czech exporting companies
11/2006	Michel Juillard Ondřej Kameník Michael Kumhof Douglas Laxton	Measures of potential output from an estimated DSGE model of the United States
12/2006	Richard Disney Sarah Bridges John Gathergood	<i>Housing wealth and household indebtedness: Is there a household 'financial accelerator'?</i>
13/2006	Frederic S. Mishkin Klaus Schmidt- Hebbel	Does inflation targeting make a difference?
1/2007	Fabrizio Coricelli Roman Horváth	Disaggregate evidence on the Czech Republic
2/2007	Ian Babetskii Nauro F. Campos Ian Babetskii	Does reform work? An econometric examination of the reform-growth puzzle Measuring and explaining inflation persistence:
3/2007	Jan Brůha Jiří Podpiera Stanislav Polák	The convergence of a transition economy: The case of the Czech Republic
4/2007	Roman Horváth	<i>The time-varying policy neutral rate in real time: A predictor for future inflation?</i>
5/2007	Jiří Podpiera Laurent Weill	Bad luck or bad management? Emerging banking market experience
	Laurent Weill Franziska Schobert	

13/2005	Robert F. Engle Jose Gonzalo Rangel	The spline GARCH model for unconditional volatility and its global macroeconomic causes
12/2005	Jaromír Beneš Tibor Hlédik Michael Kumhof David Vávra	An economy in transition and DSGE: What the Czech national bank's new projection model needs
11/2005	Marek Hlaváček Michael Koňák Josef Čada	The application of structured feedforward neural networks to the modelling of daily series of currency in circulation
10/2005	Ondřej Kameník	Solving SDGE models: A new algorithm for the sylvester equation
9/2005	Roman Šustek	Plant-level nonconvexities and the monetary transmission mechanism
8/2005	Roman Horváth	Exchange rate variability, pressures and optimum currency area criteria: Implications for the central and eastern european countries
7/2005	Balázs Égert Luboš Komárek	Foreign exchange interventions and interest rate policy in the Czech Republic: Hand in glove?
6/2005	Anca Podpiera Jiří Podpiera	Deteriorating cost efficiency in commercial banks signals an increasing risk of failure
5/2005	Luboš Komárek Martin Melecký	The behavioural equilibrium exchange rate of the Czech koruna
4/2005	Kateřina Arnoštová Jaromír Hurník	<i>The monetary transmission mechanism in the Czech Republic (evidence from VAR analysis)</i>
3/2005	Vladimír Benáček Jiří Podpiera Ladislav Prokop	Determining factors of Czech foreign trade: A cross-section time series perspective
2/2005	Kamil Galuščák Daniel Münich	Structural and cyclical unemployment: What can we derive from the matching function?
1/2005	Ivan Babouček Martin Jančar	<i>Effects of macroeconomic shocks to the quality of the aggregate loan portfolio</i>
10/2004	Aleš Bulíř Kateřina Šmídková	Exchange rates in the new EU accession countries: What have we learned from the forerunners
9/2004	Martin Cincibuch Jiří Podpiera	Beyond Balassa-Samuelson: Real appreciation in tradables in transition countries
8/2004	Jaromír Beneš David Vávra	Eigenvalue decomposition of time series with application to the Czech business cycle
7/2004	Vladislav Flek, ed.	Anatomy of the Czech labour market: From over-employment to under-employment in ten years?
6/2004	Narcisa Kadlčáková Joerg Keplinger	Credit risk and bank lending in the Czech Republic
5/2004	Petr Král	Identification and measurement of relationships concerning inflow of FDI: The case of the Czech Republic
4/2004	Jiří Podpiera	Consumers, consumer prices and the Czech business cycle identification
3/2004	Anca Pruteanu	The role of banks in the Czech monetary policy transmission mechanism
2/2004	Ian Babetskii	EU enlargement and endogeneity of some OCA criteria:

		Evidence from the CEECs
1/2004	Alexis Derviz Jiří Podpiera	<i>Predicting bank CAMELS and S&P ratings: The case of the Czech Republic</i>

CNB RESEARCH AND POLICY NOTES				
1/2008	Nicos Christodoulakis	<i>Ten years of EMU: Convergence, divergence and new policy prioritie</i>		
2/2007	Carl E. Walsh	Inflation targeting and the role of real objectives		
1/2007	Vojtěch Benda Luboš Růžička	Short-term forecasting methods based on the LEI approach: The case of the Czech Republic		
2/2006	Garry J. Schinasi	Private finance and public policy		
1/2006	Ondřej Schneider	The EU budget dispute – A blessing in disguise?		
5/2005	Jan Stráský	<i>Optimal forward-looking policy rules in the quarterly projection model of the Czech National Bank</i>		
4/2005	Vít Bárta	Fulfilment of the Maastricht inflation criterion by the Czech Republic: Potential costs and policy options		
3/2005	Helena Sůvová Eva Kozelková David Zeman Jaroslava Bauerová	Eligibility of external credit assessment institutions		
2/2005	Martin Čihák Jaroslav Heřmánek	Stress testing the Czech banking system: Where are we? Where are we going?		
1/2005	David Navrátil Viktor Kotlán	The CNB's policy decisions – Are they priced in by the markets?		
4/2004	Aleš Bulíř	External and fiscal sustainability of the Czech economy: A quick look through the IMF's night-vision goggles		
3/2004	Martin Čihák	Designing stress tests for the Czech banking system		
2/2004	Martin Čihák	Stress testing: A review of key concepts		
1/2004	Tomáš Holub	Foreign exchange interventions under inflation targeting: The Czech experience		

CNB ECONOMIC RESEARCH BULLETIN

November 2010	Wage adjustment in Europe
May 2010	Ten years of economic research in the CNB
November 2009	Financial and global stability issues
May 2009	Evaluation of the fulfilment of the CNB's inflation targets 1998–2007
December 2008	Inflation targeting and DSGE models
April 2008	Ten years of inflation targeting
December 2007	Fiscal policy and its sustainability
August 2007	Financial stability in a transforming economy
November 2006	ERM II and euro adoption
August 2006	Research priorities and central banks

November 2005	Financial stability
May 2005	Potential output
October 2004	Fiscal issues
May 2004	Inflation targeting
December 2003	Equilibrium exchange rate

Czech National Bank Economic Research Department Na Příkopě 28, 115 03 Praha 1 Czech Republic phone: +420 2 244 12 321 fax: +420 2 244 14 278 http://www.cnb.cz e-mail: research@cnb.cz ISSN 1803-7070